Author’s response to reviews

Title: The three CYBA variants (rs4673, rs1049254 and rs1049255) are benign: new evidence from a patient with CGD

Authors:

Jinqiao Sun (jinqiaosun@sina.com)
Min Wen (mwen14@fudan.edu.cn)
Ying Wang (joyce_0322@163.com)
Danru Liu (danruliu1981@163.com)
Wenjing Ying (eveline1111@163.com)
Xiaochuan Wang (xchwang@shmu.edu.cn)

Version: 3 Date: 27 Oct 2017

Author’s response to reviews:

Dear Prof. Menard,

Many thanks for your decision letter dated on Oct 24. We hereby resubmit our manuscript entitled “The three CYBA variants (rs4673, rs1049254 and rs1049255) are benign: new evidence from a patient with CGD” (MGTC-D-17-00023R3) for consideration for publication in BMC Medical Genetics.

Thank you for your comments. We have revised the manuscript. I hope you enjoy reading.

1. We revised the Consent to participate section and moved it to Declarations (see line 8-12, page 10).

2. We revised the headings in the Abstract (see line 9-11, page 2).
3. We revised the headings in the main text (see line 1, page 4; line 3, page 5; line 17, page 9).

4. We revised the Contributions of authors (see line 7-11, page 11).

5. We revised the Declarations (see page 10-11).

6. We moved the Abbreviations to the Declarations section (see line 2-7, page 10).

7. We only keep one set of Competing Interests sections (see line 1-3, page 11) and Consent for publication sections (see line 13-15, page 10).

Thank you for your work again.

Best Regards,

Xiaochuan Wang